Author’s response to reviews

Title: Test–retest reliability of tip, key, and palmar pinch force sense in healthy adults

Authors:

Lin Li (20160055@ruc.edu.cn)
Yanxia Li (lmm333_0@163.com)
Changhong Wu (17145575@qq.com)
Xinyan Zhang (xinyanzhang_ruc@163.com)

Version: 1 Date: 16 Jun 2019

Author’s response to reviews:

Dear Amanpreet Athwal,

Thank you for considering the revised version of our manuscript, " Test–retest reliability of tip, key, and palmar pinch force sense in healthy adults ", submitted to BMC Musculoskeletal Disorders. We are thankful to the Editor for pointing out some important modifications needed in the report. We have thoughtfully taken into account these comments.

1. " All study participants gave consent for their identifying image in Figure 1 to be published in this study." was added into Consent for Publication.

2. " The funding body had no role in the design of the study, collection, analysis, or interpretation of data or in writing the manuscript. " was added into Funding.

3. Order my Declarations section in the following way:
   Ethics approval and consent to participate
   Consent for publication
   Availability of data and material
   Competing interests
   Funding
   Authors' contributions
   Acknowledgements

4. My manuscript was edited by engaging a native speaker.

We hope that these changes fulfil the requirements to make the manuscript acceptable for publication in BMC Musculoskeletal Disorders.

Looking forward to hearing from you soon.

Sincerely yours,
Yanxia Li